Author’s response to reviews

Title: The clinical practice guideline Palliative care for children and other strategies to enhance shared decision-making in pediatric palliative care; pediatricians’ critical reflections

Authors:

D. Dreesens (dunja.dreesens@maastrichtuniversity.nl; dunja.dreesens@gmail.com)

Lotte Veul (lotte.veul@gmail.com)

Jonne Westermann (j.westermann@ggzcentraal.nl)

Nicole Wijnands (nwijnands@outlook.com)

Leontien Kremer (l.c.kremer@amc.uva.nl)

Trudy van der Weijden (trudy.vanderweijden@maastrichtuniversity.nl)

Eduard Verhagen (a.a.e.verhagen@umcgroningen.nl)

Version: 4 Date: 20 Nov 2019

Author’s response to reviews:

The editor requested to specify in the manuscript the contributions of two authors; this has been specified in the enclosed version of the manuscript, p.21. Furthermore, it was requested to only upload the files necessary for publication. The files with the tables (also included in the manuscript), the SRQR checklist and revision note have not been uploaded this time.